Author's response to reviews

Title: Acquired Genetic Alterations in Tumor Cells Dictate the Development of High-Risk Neuroblastoma and Clinical Outcomes

Authors:

Faizan H Khan (Faizan-Khan@ouhsc.edu)
Vijayabaskar Pandian (Vijayabaskar-Pandian@ouhsc.edu)
SatishKumar Ramraj (Satish-Ramraj@ouhsc.edu)
Mohan Mohan Natarajan (natarajan@uthscsa.edu)
Sheeja Aravindan (Sheeja-Aravindan@ouhsc.edu)
Terence S Herman (Terence-Herman@ouhsc.edu)
Natarajan Aravindan (naravind@ouhsc.edu)

Version: 2
Date: 4 December 2014

Author's response to reviews:

1. Line number were added to the manuscript.
2. Now it is mentioned in manuscript that human data used were obtained from public database (http://r2.amc.nl).